Author's response to reviews

Title: Drug Utilization Evaluation of Cefepime in the First Affiliated Hospital of Bengbu Medical College: A Retrospective and Prospective Analysis

Authors:
Shi Qingping (sir_shi@126.com)
Ding Feng (958387264@qq.com)
Sang Ran (13855288331@163.com)
Liu Yan (1055237127@qq.com)
Yuan Haoyu (302989860@qq.com)
Yu Meiling (yumeiling@sohu.com)

Version: 9 Date: 22 March 2013

Author's response to reviews: see over
Dear Dr. Harris,

I was very glad to receive your letter. Thank you very much for your patience and the requirement of some final clarifications on the ethical approval and consent sought for our work. I would like to provide this cover letter the concerns raised.

Reviewer: Editors

We would like to request that you modify the text regarding ethical approval of your work to read:

This study was approved by the Department of Health of Anhui Province in China and the Ethics Committee of The First Affiliated Hospital of Bengbu Medical College.

The need for informed consent for this study was waived by the ethics committee.

The revised section was marked in blue in the manuscript.

The above documents have been submitted in electronic version to BMC Infectious Diseases. If you have any questions or queries, please write to me by the email given below. I am looking forward to acknowledge the receipt of this article by email.

Thank you very much.

Yours sincerely,
Qingping Shi
Department of Pharmacy,
The First Affiliated Hospital of Bengbu Medical College,
Bengbu 233004, China.
Email: sir_shi@126.com
Tel: +86-552-3086246.